Author’s response to reviews

Title: Equity-oriented toolkit for health technology assessment and knowledge translation: application to scaling up of training and education for health workers

Authors:

Erin Ueffing (erin.ueffing@uottawa.ca)
Peter Tugwell (elacasse@uottawa.ca)
Janet Hatcher Roberts (jroberts@csih.org)
Peter Walker (pwalker@uottawa.ca)
Nadia Hamel (nadiah@uottawa.ca)
Vivian Welch (vivian.welch@uottawa.ca)

Version: 4 Date: 26 July 2009

Author’s response to reviews:

To Whom It May Concern:

We have made the following minor changes to our manuscript:

Page 3, para 2: Capitalized first letter of each word in “World Health Report”

Page 4, second last paragraph: Changed quotation marks from straight to typographic around Needs-Based Technology Assessment: Exploring Global Interfaces

Page 9, Acknowledgements: We have reworded the last sentence of the Acknowledgements to include Janet Clevenstine.

Page 11, References 18-24: Added issue numbers and article numbers for each of the Cochrane reviews.

Page 11, Reference 18: Updated with full review.

Thank you very much.

Best regards,

Erin Ueffing